Author’s response to reviews

Title: Growth Data and Tumour Risk of 32 Chinese children and adolescents with 45, X/46, XY mosaicism

Authors:

Lili Pan (panrilier@163.com)
Zhe Su (Su_zhe@126.com)
Jianming Song (jamie898@163.com)
Wanhua Xu (13510332826@163.com)
Xia Liu (1136886046@qq.com)
Longjiang Zhang (zljl8989@126.com)
Shoulin Li (lishoulin@126.com)

Version: 3 Date: 18 Apr 2019

Author’s response to reviews:

Reply to the editor’s comments:

1. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos).

If you wish to publish Table 1, with its list of clinical features alongside your manuscript, you will need to have detailed that patients and/or their legal proxies provided written informed consent for the publication of potentially identifiable information. If you were/are unable to acquire written informed consent for publication, you may instead remove Table 1 from your manuscript and provide the de-identified data upon reasonable request from future readers.

The statement regarding authors’ approval of the final manuscript should instead be included in the Authors’ contributions section of the Declarations.

Reply: Thank you for your comment. We have removed table 1. We will provide the de-identified data upon reasonable request from future readers.

Literature search results section, line 1, page 9: Table 2 was changed to Table 1.
Ethics approval and consent to participate section, line 17, page 11: We have removed 'The patients provided informed consent, and' and 'Written informed consent was obtained from a parent or guardian for each participant.'

Authors’ contributions section, line 61, page 11: We have added 'All authors read and approved the final manuscript.'